Author's response to reviews

Title: Unusual insidious spinal accessory nerve palsy: A case report

Authors:

Ioannis N Charopoulos (jcharopoylos@yahoo.com)
Nikolas Hadjinicolaou (Nickydoc2000@hotmail.com)
Ioannis Aktselis (Aktselis@hotmail.com)
George P Lyritis (glyritis@mycosmos.gr)
Nikolaos Papaioannou (jcharopoylos@yahoo.com)
Constantinos Kokorogiannis (ckokorogiannis@hotmail.com)

Version: 2 Date: 26 January 2010

Author's response to reviews:

Firstly i corrected the authorship sequence both in the submission system and manuscript. Moreover, i removed the ??? from the submission abstract. Concerning the comment about the nationality and ethnicity of the patient, in Greece the nationality and ethnicity is the same since the homogeneity of Greek population. Finally, i removed as it was pointed out the received and accepted date sections from the title page.